# RESEARCH





Socio-ecological factors associated with probable depression among pregnant and parenting adolescent girls: findings from a cross-sectional study in Burkina Faso and Malawi

Anthony Idowu Ajayi<sup>1\*</sup>, Elita Chamdimba<sup>2</sup>, Nathalie Sawadogo<sup>3</sup>, Nyawira Gitahi<sup>4</sup>, Abdoul Moumini Tarnagda<sup>3</sup>, Abdoul Kader Ilboudo<sup>3</sup>, Alister Munthali<sup>2</sup>, Chrissie Thakwalakwa<sup>2</sup>, Emmanuel Oloche Otukpa<sup>1</sup>, Boniface Ayanbekongshie Ushie<sup>1</sup> and Caroline W. Kabiru<sup>1</sup>

## Abstract

**Background** Pregnant and parenting adolescent girls are at risk of poor mental health because of stigma and social exclusion. Despite one in four girls starting childbearing by the age of 19 in Africa, no study, to the best of our knowledge, has examined the multi-layered factors (individual, family, friends, and neighborhood-related factors) associated with depressive symptoms among pregnant and parenting girls in Africa. Our study contributes to addressing this gap by examining the socio-ecological factors associated with depression symptoms among pregnant and parenting adolescent girls.

**Methods** Our study adopted a cross-sectional design. Between March and September 2021, we interviewed 980 pregnant and parenting adolescent girls in Ouagadougou, Burkina Faso, and 669 in Blantyre, Malawi. We recruited pregnant and parenting adolescent girls in randomly selected urban and rural enumeration areas in Burkina Faso (n = 71) and Malawi (n = 66). We assessed depressive symptoms using the Patient Health Questionnaire (PHQ-9), which generated an overall score of 27. We considered a score of 10 or more as probable depression. We also obtained information on individual, family, friends, and neighborhood characteristics. We employed logistic regression models to examine the significant factors associated with probable depression among pregnant and parenting adolescent girls.

**Results** The prevalence of probable depression was 18.8% and 14.5% in Burkina Faso and Malawi, respectively. At the individual level, having secondary education was significantly associated with a lower likelihood of probable depression in Malawi (AOR: 0.47; 95% CI 0.27–0.82) but not in Burkina Faso. At the family level, denying paternity (AOR: 3.14; 95% CI 1.34–7.11 in Malawi) and no parental support (AOR: 2.08; 95% CI 1.22–3.55 in Burkina Faso) were associated with higher odds of probable depression. At the community level, perceived neighborhood safety was associated with a lower likelihood of probable depression in Malawi (AOR: 0.74; 95% CI 0.61–0.89) and Burkina Faso (AOR: 0.81;

\*Correspondence: Anthony Idowu Ajayi aajayi@aphrc.org Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

95% CI 0.73–0.90). Having a safety net within the community was associated with lower odds of probable depression in Burkina Faso (AOR: 0.87; 95% CI 0.78–0.96) but not in Malawi.

**Conclusion** Depressive symptoms are common among pregnant and parenting adolescents, suggesting the need to screen them regularly for depression during antenatal and postnatal visits. Factors associated with depression among pregnant and parenting girls operate at multiple levels suggesting a need for multilevel interventions that address all areas of vulnerabilities.

## **Plain English Summary**

Pregnant and parenting adolescents are at a higher risk of poor mental health. If unaddressed or untreated, the consequences may be dire throughout an individual's life. About one in four girls start childbearing before their 19th birthday in Africa. Most of these pregnancies are unintended or unplanned and occur outside wedlock, bringing untoward disruption to every facet of the girl's life. The burden of early parenthood and the stigma of starting childbearing too early and outside wedlock can affect the mental wellness of girls. Yet only a few studies have focused on understanding the burden and drivers of poor mental health among pregnant and parenting girls. Our study contributes to the research on depression among pregnant and parenting girls, highlighting its burden and individual, family, and community drivers. We conducted this study in Ouagadougou, Burkina Faso, and Blantyre, Malawi, between March and September 2021. A total of 980 and 669 pregnant and parenting girls were recruited into this study in Burkina and Malawi. Our analysis shows that 18.8% and 14.5% of pregnant and parenting girls in Burkina Faso and Malawi, respectively, had probable depression. Depression symptoms was higher when girls were older than 17 years, suffered violence from their partners, their partners denied paternity, received no support from their partners and parents, lived in violent neighborhoods, and had no support system in the community. We conclude that depression is high among pregnant and parenting girls. It necessitates screening them for depression during antenatal and postnatal care visits. To effectively address depression among these girls, interventions must canvass support at the family and community levels.

**Keywords** Mental health, Depression, Pregnant and parenting adolescents, Adolescent mothers, Socio-ecological model

## Introduction

Mental health problems significantly contribute to the disease burden among adolescents and young adults globally [1]. Half of all mental health conditions begin in adolescence, and one in six people aged 10–19 years suffer from some form of mental disorder [2]. Depression and suicide are among the leading causes of death and disabilities among adolescents [1]. However, most mental conditions are undiagnosed and untreated, especially in resource-poor settings [3]. Untreated mental conditions have detrimental consequences across the life course, damaging physical health and overall wellness and robbing individuals of opportunities to lead fulfilling lives as adults [3]. Untreated mental health problems among adolescents are also linked to low educational achievement, unemployment, substance abuse, risky behaviors, crime, and self-harm, leading to a heightened lifetime risk of morbidity and mortality [2].

The reasons why adolescents are susceptible to poor mental health outcomes are multifaceted. Childhood abuse, family, school, and neighborhood violence, poverty, loss and grief, social exclusion, early unintended pregnancy, and chronic diseases are among the risk factors for poor mental health outcomes among adolescents [3–6]. Vulnerable adolescents, including those pregnant, living with HIV or disability, orphaned, trafficked, and forced into sex work, are disproportionately susceptible to poor mental health, including depression, anxiety, suicide ideation, self-harm, and traumatic stress<sup>[7]</sup>. There is evidence that pregnant and parenting adolescent (PPA) girls are disproportionately exposed to physical abuse, more stressful life events, and postpartum depressive symptoms than adult women [4, 8, 9]. For example, the risk of postpartum depression is doubled in adolescent mothers compared to their non-parenting peers and adult women [10]. Elevated levels of mental health challenges among PPA girls are a function of their age and other social disadvantages and adversities that precede pregnancy [5, 8].

Some of the disadvantages that make adolescent girls vulnerable to early and unintended pregnancies also elevate their risk of poor mental health. Studies from the US show that the childhoods of adolescent mothers from resource-poor communities and poor households are characterized by social inequality, chronic stress, violence, food insecurity, and poverty, all of which increase their risk of experiencing rape and sexual violence and unintended pregnancy. Some became pregnant due to sexual abuse, a risk factor for poor mental health [5]. When they become mothers, their adversities become compounded by the stress of parenting, poverty, lack of resources to care for their babies, and the stigma of early childbearing outside wedlock, especially when the sexual partners deny responsibility and offer no support [11, 12].

Much of the literature on adolescent mental health is concentrated in the Global North [11, 12]. More recently, there has been an increased focus on the mental health of adolescents living with HIV in Africa [13-17]. However, limited research, program interventions, and policies have addressed pregnant and parenting adolescents' mental health and wellness. Even though Africa has the highest prevalence of adolescent childbearing globally, interventions to improve the mental health and wellbeing of adolescent mothers are concentrated in the global north [18]. Diagnosing and treating mental distress in many African countries is limited and under-resourced. Most people suffering from poor mental health lack access to much-needed care. While a few studies have examined the factors associated with depression symptoms among PPA girls in sub-Saharan Africa [19, 20], none to the best of our knowledge has comprehensively examined the role of individual, family, friends, and neighborhood-related factors associated with depressive symptoms among pregnant and parenting girls. Our study addresses this gap by examining the socio-ecological factors associated with depression symptoms among PPA girls in rural and urban areas in two districts of Burkina Faso and Malawi.

## **Theoretical underpinning**

This study drew from the socio-ecological model to understand the multi-layered factors associated with probable depression among pregnant and parenting adolescents [21-24]. We argue that factors predisposing PPA girls to probable depression are complex and multi-layered. The interplay of individual, family, friend, and community level factors combine to expose girls to poor or good mental health [22, 24]. When girls become pregnant, often outside wedlock, they face social exclusion at home, from friends and in the community, which could culminate in poor mental health [9]. Early and unintended pregnancy constitutes a barrier to girls' education and economic empowerment [25]. It exposes them to physical and emotional abuse from their families, partners, and community [26]. Often, their friends judge, mock, isolate, and gossip about them, which could lead to poor mental health [20]. Because they are very young mothers, they need much support to navigate parenting, support that is often not available to them [27].

Sometimes, their partners deny paternity and refuse to support them, which make them vulnerable to poor mental health [28, 29]. Previous studies have also shown that community or neighborhood-level factors are relevant in understanding poor mental health among pregnant and parenting adolescents [30–32]. Girls who live in poor neighborhoods where crime and violence are rampant have poorer mental health than their counterparts [31, 32]. Without support from family, partners, friends, and community, adolescent mothers may be unable to deal with the compounded challenges of motherhood.

## Methods

## Study design and setting

We analyzed data from a cross-sectional study that examined pregnant and parenting adolescents' lived experiences in Burkina Faso in West Africa and Malawi in Southern Africa. We selected these countries purposively because they have some of the highest adolescent childbearing rates in the world—136 births per 1000 girls in Malawi and 132 births per 1000 girls in Burkina Faso [33]. The choice of Blantyre and Ouagadougou was convenient given the data was collected during COVID-19 and adolescent childbearing is common in all regions of the countries.

## Sampling procedure

We used a two-stage probability sampling approach to recruit pregnant and parenting girls into the study. The first stage involved random sampling of urban and rural enumeration areas (EAs). Overall, 66 EAs (rural=26 and urban=40) were sampled in Blantyre, Malawi, and 71 EAs (rural = 29 and urban = 42) were selected in the Central region that includes Ouagadougou and six rural municipalities (Pabré, Loumbila, Tanghin Dassouri, Saaba, Komsilga and Koubri) around the town in Burkina Faso. In the second stage, we conducted a household listing in the sampled EAs to identify households with eligible pregnant and parenting adolescent girls. To be eligible to participate, girls had to be aged 10-19 years, ever been pregnant, pregnant or have a biological child, and be mentally capable of responding to the questions in English, French, Chichewa, or the local languages. All identified adolescents were selected per household. In total, we successfully interviewed 980 and 669 adolescent girls in Ouagadougou and Blantyre, giving a response rate of 96.2% (n=980/1019) and 98.5% (n=669/679), respectively.

## Data collection procedures

Data were collected using a structured questionnaire adapted from existing instruments, such as the Global Early Adolescent Study tools [34], and Patient Health Questionnaire-9 (PHQ-9) [35–37]. We administered the study questionnaires in the English, French and national languages: Chichewa in Malawi and Moore and Dioula in Burkina Faso. The questionnaire was piloted in both study settings, and identified issues were corrected before we administered the final version. Data collection took place between March and September 2021. Trained enumerators used Android-powered tablets and smartphones with SurveyCTO to collect and upload data to a secured server. The research teams were managed by supervisors who monitored the data collection process by cross-checking for errors daily before syncing the data onto the server.

### Main outcome

We used the Patient Health Questionnaire-9 (PHQ-9) to determine depression symptoms among PPA girls [35-37]. The PHQ-9 has over 75% sensitivity and specificity for screening depression in adolescents [38] and is widely used and validated in African settings [39-42]. The PHQ-9 is a nine-item self-assessment instrument with response options of 0 (not at all); 1 (several days), 2 (more than half the days), and 3 (nearly every day). We asked each respondent how often they have been bothered by the following problems over the last two weeks: having little interest or pleasure in doing things; feeling down or hopeless; having insomnia or oversleeping; fatigue; loss of appetite or over-eating, having low self-esteem; trouble concentrating; restlessness or slowness; and negative selfthoughts, including self-harm. Scores of 10 to 27 are further grouped as probable depression and zero to nine are grouped as no probable depression. The questions have a high degree of internal consistency in our study with a Cronbach's alpha coefficient of 0.84 and 0.85 in Malawi and Burkina Faso.

## Covariates

We included three levels of covariates, individual, family and friends, and community levels. At the individual level, we included age (13–16, 17, 18, and 19 years), marital status (married/cohabiting, separated/divorced, and single), education level (no education/primary and secondary), marital status at the time of first pregnancy (single, engaged and married), personal reaction to the pregnancy (scared, happy, disappointed), and birth status (ever given birth/has at least one child and still pregnant/never given birth). At the family and friend level, we included exposure to intimate partner violence, parent's reactions to pregnancy (happy upset and neutral), friends' reactions to pregnancy (happy upset and neutral), partner's reactions to pregnancy (happy, denied, upset and neutral), parental support (good, fair, poor/no support), friend's support (good, fair, poor/no support), and partner's support (good, fair, poor/no support). Exposure to intimate partner violence (IPV) was generated using the Demographic and Health Survey IPV module and any experience of sexual, emotional and physical violence was grouped as "Yes" and no experience was grouped as "No".

At the community level, we included perceived neighborhood safety and social assets and safety nets. Seven questions were used to assess neighborhood safety, and these questions have a moderate internal consistency and validity with an alpha coefficient of 0.42. Each participant responded by indicating the degree of agreement or disagreement to questions about feeling safe walking around their community during the day and after dark, feeling scared of being raped, and having previously been robbed or indecently touched. Social capital was measured using a scale comprising seven items (alpha coefficient of 0.66). The questions explored if PPA girls agree or disagree with statements on having someone they can rely on for child support, advice, loan, psychosocial support, or help.

## Statistical analysis

We used SPSS (version 25) for data cleaning and Stata (version 15 for Windows) [43] for all the statistical analyses. The survey datasets were weighted to correct for over or under sampling of some areas. Descriptive statistics were produced using means, standard deviations, frequencies, and percentages to present the demographic characteristics of the respondents and observations of probable depression and no depression. We fitted adjusted and unadjusted models to examine individual, family/friend, and community levels correlate of probable depression among pregnant and parenting girls. A p-value < 0.05 was considered statistically significant, and 95% confidence intervals were estimated. Model 1 is the unadjusted model showing the effect of each variable on the study outcome, and Model 2 is the adjusted model showing the effect of each variable after controlling for covariates. We dropped living with both parents and orphanhood status from the multivariable analysis because of multicollinearity. Living with both parents and orphanhood status significantly correlate with parental support.

### **Ethical considerations**

The Research Ethics Committee of the University of Malawi (UNIMAREC) and Burkina Faso's Ministry of Health Ethics Committee for Health Research approved the study protocol. Interviews were conducted in private spaces, and participants' rights to confidentiality and anonymity were respected during and after data collection. Married pregnant and parenting adolescents provided oral and written informed consent after the research assistants explained the aim of the study and the use of data to them. Parents and guardians of single pregnant girls gave informed consent, and their wards assented to the study. We observed all IRB's ethics guidelines for research on human subjects.

## Results

## **Descriptive findings**

The average age of pregnant and parenting girls in Malawi and Burkina Faso was  $18.9 \pm 1.2$  years and  $18.4\pm0.9$  years, respectively. Table 1 presents the weighted individual, family, and community characteristics of respondents. In both Malawi (66.9%) and Burkina Faso (80.6%), most PPA girls were aged 18 to 19. More respondents had ever worked for pay in Burkina Faso (36.9%) compared to Malawi (28.7%). Similarly, more respondents in Burkina Faso (84.2%) than in Malawi (51.0%) were not living with both parents. While over three-quarters of PPA girls in Burkina Faso (75.6%) were married/cohabiting, less than half in Malawi (45.8%) were married. In contrast, a higher proportion of PPA girls in Malawi (63.3%) described parental support as good compared to those from Burkina Faso (47.2%). However, more PPA girls from Burkina Faso (75.2%) described the partners/spousal support as good compared to those from Malawi (61.9%). More respondents got pregnant outside of formal union in Malawi (80.6%) than in Burkina Faso (56.4%).

## Probable depression prevalence

Table 2 presents the prevalence of probable depression by respondents' individual, family, and communitylevel characteristics. The prevalence of probable depression was 14.5% and 18.8% in Malawi and Burkina Faso, respectively. The prevalence of probable depression was higher among single PPA girls compared to married adolescents in both Malawi (single 17.7% vs. married 9.8%) and Burkina Faso (single (25.4% vs. married 16.6%). Probable depression prevalence was lower among those who received good support from parents, friends, and partners than those who received poor or no support in both countries (Table 2). Similarly, probable depression prevalence was lower among respondents who were married before getting pregnant than those single when they got pregnant in both study sites. Probable depression prevalence was higher among respondents who reported intimate partner violence than those who did not in both study sites.

### Multivariable findings

Our models examined individual, family and friend, and community-level factors associated with probable depression among pregnant and parenting adolescent girls. The individual-level factors associated with probable depression in the unadjusted model were age, marital status, educational level, and birth status (Table 3). After adjusting for covariates, increasing age was associated with higher odds of probable depression in both Malawi and Burkina Faso. However, the effect size reached a statistically significant level only in Malawi. While being single and separated was significantly associated with a higher likelihood of probable depression in the unadjusted model, the effect size did not reach a significant level after controlling for relevant covariates. Having secondary education was significantly associated with a lower likelihood of probable depression in Malawi but not in Burkina Faso. Parenting girls were more likely to report lower levels of probable depression compared to those currently pregnant in Burkina Faso but not in Malawi.

The family and friend-related factors significantly associated with probable depression in the unadjusted model were IPV, parental, partner, and friends' reactions to pregnancy, and their level of support. However, only exposure to IPV, denial of paternity, and parental and partner support reached a significant level in the adjusted model. Pregnant and parenting girls who reported experiencing IPV were more likely to report depression symptoms than those who did not in both study settings. In Malawi and Burkina Faso, PPA girls who reported that their partner denied paternity were more likely to report probable depression than those whose partners were happy with their pregnancy. Similarly, those who described the support they received from their parents and partners as poor, or fair were more likely to report probable depression than those who received good support (Table 3).

At the community level, perceived neighborhood safety and having support and a safety net within the community were protective against depression symptoms. Pregnant and parenting adolescent girls who perceived their community as safe were less likely to report probable depression in Malawi and Burkina Faso. Safety nets and support within the community were associated with lower odds of probable depression in Burkina Faso but not in Malawi.

## Discussion

Research on depression among pregnant and parenting adolescents is limited in Africa. Such limited focus on depression among this vulnerable group has implications for designing appropriate interventions or advocacy for better access to care. Our study contributes to the evolving literature on the topic by using the widely used socio-ecological model [21, 23, 24] to examine the factors associated with probable depression among PPA girls in Burkina-Faso and Malawi. We found a 14.5% and

# Table 1 Weighted individual, family and friend, and community characteristics of respondents

Variables	Malawi Weighted sample		Burkina Faso Weighted sample	
	Frequency	Percent	Frequency	Percent
Residence				
Rural	275	38.2	398	19.1
Urban	394	61.8	582	80.9
Age				
13–16	87	13.3	71	6.9
17	133	19.9	127	12.5
18	199	29.8	313	32.1
19	249	37.1	469	48.5
Education level				
Primary or less	440	65.4	526	53.0
Secondary	229	34.6	454	47.0
Ever worked for pay	227	5		
Yes	191	28.7	344	36.9
No	478	71.3	636	63.1
Orphanhood status	1/0	71.5	050	05.1
Double oroban	30	5.8	54	61
Single orphan	1.81	27.1	208	20.7
Non-orphan	101	67.1	718	20.7 73 0
Living with both parents	449	07.1	710	15.2
Not living with both parents	241	50.0	022	04.0
Not inving with both parents	170	20.9	632	64.2
Living with both parents	170	20.5	05	0.0
Living with both parents	152	22.0	60	9.1
Married	207	4E 0	761	75.6
Married	307	43.0	701	75.0
Separated	79	11.8	59	59
Single Bisth status	283	42.3	160	17.8
Birth Status	144	21.4	207	20.0
Ever given birth/has at least one child	144	21.4	297	28.8
Still pregnant/never gave birth	525	/8.6	686	/1.2
Personal reaction to pregnancy	100	27.2	225	265
Scared	183	27.3	335	36.5
нарру	187	27.8	435	42.5
Disappointed	299	44.9	210	21.0
Marital status at the time of pregnancy	520	00.7	527	54.4
Single	539	80.7	537	56.4
Engaged	15	2.3	38	3.5
Married	115	17.0	405	40.2
Parental support	105	62.2		17.0
Good	425	63.3	445	47.2
Fair	136	20.5	3/1	37.0
Poor	33	4.9	/5	7.1
No support	/5	11.2	89	8.7
Friends support				
Good	155	22.9	250	25.0
Fair	137	21.0	433	45.8
Poor	83	12.4	61	6.4
No support	294	43.7	236	22.8

## Table 1 (continued)

Social assets and safety net (median)

	<b>.</b>		D 11 F	
Variables	Malawi		Burkina Faso	
	Weighted sample		Weighted sample	
	Frequency	Percent	Frequency	Percent
Partner support				
Good	415	61.9	741	75.2
Fair	108	16.4	142	14.1
Poor	48	7.2	96	10.7
No support	98	14.5	1	0.05
Neighborhood safety (median)	669	6	980	5

669

4

\* Median computed for "neighborhood safety" and "social assets and safety net'

18.8% prevalence of probable depression in Malawi and Burkina Faso, respectively. The prevalence of probable depression in the two study settings is consistent with previous studies reporting a prevalence of 13% in Zimbabwe [44] and 16% in South Africa [45]. However, studies among pregnant adolescent girls in Rwanda (48%) [46] and Kenya (33% and 52%) [47, 48] using the Edinburgh Postnatal Depression Screen tool found a far higher prevalence. The varying prevalence of probable depression is due to some differences in the periods of pregnancy and maternity taken into account (many studies focus on the antenatal or postpartum periods) and the use of different screening tools [49].

Factors associated with probable depression in both study settings are broadly similar, but a few differences exist. While age and education were the only individuallevel factors found statistically significant in Malawi, only birth status was significantly associated with probable depression in Burkina Faso. Increasing age was associated with high levels of possible depression, while having a secondary education was protective. Our finding on the link between age and depression is consistent with a previous study in Tanzania [50]. The plausible explanation for the association between age and probable depression is that increasing age is linked to adult responsibility [51]. Previous studies have shown the protective benefit of having a higher education level against depression [51, 52]. The protective effect of education against depression in Malawi suggests that girls with some secondary education could be more hopeful for a future career than girls with no education or only primary education. In Malawi, higher education is linked to better future career prospects than no education or having only primary education.

Consistent with previous studies that reported a higher prevalence of depression among pregnant adolescents [46–48], our Burkina Faso result shows that girls with one or more birth were less likely to show symptoms of depression compared to those currently pregnant. This group of girls may have had time to deal with the disappointment of early and unintended pregnancy and devised a coping strategy. Our data shows that some got married and received good support from their partners, which could contribute to helping them overcome the disappointment and fear of early childbearing. Because communities frown at out-of-wedlock pregnancies and exclude such girls, child marriage is often used as a way to restore family honor or avoid public ridicule [53, 54]. Girls who get married may, therefore, be able to overcome the stigma associated with adolescent pregnancies.

980

At the household and family level, exposure to IPV, parental support, partner support, and paternity denial were associated with probable depression. All four factors were significantly associated with probable depression in Burkina Faso, but only IPV and parental support were significant in Malawi. The finding on the association between IPV and probable depression is consistent with previous studies [55-58]. The plausible explanation is that when adolescent girls become pregnant, often outside wedlock, they voluntarily or involuntarily move out of their parent's house to live with their partners. Parental rejection makes their partner the only available form of support. If the partner abuses them physically, emotionally, and sexually, they can become hopeless or even suicidal, increasing their risk of depression.

One key finding of this study is that paternity denial is a risk factor for probable depression in PPA girls, as we found in Burkina Faso. Paternity denial has serious negative effects on girls. When a girl becomes pregnant outside wedlock, she is subjected to ridicule and shame, and it gets worse if there is no one accepting responsibility for her pregnancy. She is viewed as promiscuous, lacking honor, and excluded by friends, family, and the community. Denial of paternity means a lack of material

5

## Table 2 Depression symptoms prevalence by individual, family, and community-level characteristics

Variables	Malawi		Burkina Faso	
	Not depressed % (CI)	Probably depressed % (CI)	Not depressed % (Cl)	Probably depressed % (CI)
All	85.5 (82.5–87.9)	14.5 (12.1–17.5)	81.2 (78.3–83.7)	18.8 (16.3–21.7)
Age				
13–16	86.5 (77.9–92.1)	13.5 (7.9–22.1)	77.6 (65.0–86.6)	22.4 (13.4–35.0)
17	84.9 (77.8–90.0)	15.1 (10.0–22.2)	70.8 (61.3–78.9)	29.2 (21.1–38.7)
18	84.6 (78.9–88.9)	15.4 (11.1–21.1)	82.6 (77.5–86.7)	17.4 (13.3–22.5)
19	86.1 (81.1–89.9)	13.9 (10.1–18.9)	83.4 (79.3–86.8)	16.6 (13.2–20.7)
Education level				
Primary or less	83.3 (79.4–86.5)	16.7 (13.5–20.6)	82.6 (78.7–86.0)	17.4 (14.0-21.3)
Secondary	89.6 (85.0–92.9)	10.4 (7.1–15.0)	79.6 (75.2–83.3)	20.4 (16.7–24.8)
Marital status				
Married	90.2 (86.3–93.0)	9.8 (7.0–13.7)	83.4 (80.3-86.1)	16.6 (13.9–19.7)
Separated	78.5 (68.1–86.2)	21.5 (13.8–31.9)	73.4 (59.9–83.6)	26.6 (16.4-40.1)
Sinale	82.3 (77.4–86.3)	17.7 (13.7–22.6)	74.6 (66.7–81.1)	25.4 (18.9–33.3)
Ever worked for pay				
Yes	80.4 (74.2–85.4)	19.6 (14.6–25.8)	78.3 (73.2–83.7)	21.7 (17.3–26.8)
No	87.5 (84.1–90.3)	12.5 (9.7–15.9)	82.8 (79.4–85.8)	17.2 (14.2–20.6)
Orphanhood status				
Double orphan	79.5 (64.0-89.5)	20.5 (10.5-36.0)	73.1 (58.6-83.9)	26.9 (16.1-41.4)
Single orphan	81.4 (75.0-86.4)	18.6 (13.6–25.0)	78.0 (71.0-83.7)	22.0 (16.3–29.0)
Non-orphan	87.6 (84.2–90.4)	124 (96–158)	827 (795-856)	17 3 (14 4-20 5)
living with both parents	07.0 (01.2 90.1)	12.1 (5.0 15.0)	02.7 (79.5 03.0)	17.5 (11.1 20.5)
Not living with both parents	85 7 (81 7–88 9)	143 (111–183)	81 5 (78 4–84 3)	185 (157-216)
Living with one parent	83.8 (77.7–88.4)	16.2 (11.6-22.3)	76.6 (63.4-86.1)	23.4 (13.9–36.6)
Living with both parents	86.9 (80.7–91.3)	131 (87–193)	81 5 (71 0-88 8)	185 (112-290)
Residence	00.9 (00.7 91.9)	13.1 (0.7 19.5)	01.5 (71.0 00.0)	10.5 (11.2 25.0)
Rural	83 3 (78 3-87 3)	167 (127-217)	84 4 (80 5-87 7)	156 (123-195)
Urban	86.8 (83.1-89.8)	13.2 (10.2–16.9)	80.4 (77.0-83.4)	19.6 (16.6-23.0)
Parental support	00.0 (05.1 05.0)	15.2 (10.2 10.5)	00.1 (77.0 03.1)	19.0 (10.0 29.0)
Good	80.0 (85.6-01.6)	110(84-144)	85 3 (87 7_87 9)	1/17 (12 1_17 8)
Fair	77 5 (69 7-83 8)	22.5 (16.2–30.3)	72.8 (64.1_80.1)	27.2 (10.0-35.0)
Poor	824 (658-919)	176 (81_342)	63.5 (52.7-73.1)	365 (269-473)
No support	81.6 (71.4-88.7)	18.4 (11.3-28.6)	100.0	0.0
Friends support	01.0 (71.4 00.7)	10.4 (11.5 20.0)	100.0	0.0
Good	01 7 (86 2_05 1)	83(40-138)	800 (830_026)	110(74-161)
Fair	827 (754-881)	17 3 (11 9-24 6)	826 (784-862)	17.4 (13.8-21.6)
Poor	85.4 (76.0-91.6)	14.6 (8.4- 24.0)	60.6 (46.8-72.9)	39.4 (27.1-53.2)
Ne support	835 (780, 873)	16.5 (12.7 21.1)	75 5 (68.8, 81.1)	24.5 (18.0, 31.2)
Partpor support	05.5 (70.9-07.5)	10.5 (12.7-21.1)	75.5 (00.0-01.1)	24.3 (10.9-31.2)
Good	80.6 (86.1 02.2)	10//78 130)	867 (828 808)	133(10,2,17,2)
Good	79.0 (70.E 95.4)	10.4(7.0-13.9)	00.7 (02.0-09.0)	10.6 (10.2-17.2)
	76.9 (70.3-63.4)	21.1 (14.0-29.3)	61.5 (49.0, 73.9)	19.0 (15.5-24.0)
POOL	75.7 (01.2-60.0)	24.5 (14.0-50.0)	01.5 (40.9-72.0) 70.7 (E0.1, 20.1)	30.3 (27.2-31.1) 20.3 (10.0 40.0)
	00.2 (71.3-00.0)	19.0 (15.2–20.7)	70.7 (39.1–60.1)	29.5 (19.9–40.9)
Delonging to a social group		1(1(1))	010(7(7,04))	100(154, 222)
res	83.9 (/9.0-8/.8)	10.1 (12.2-21.0)	δI.U (/b./-84.6)	19.0 (15.4–23.3)
NU Marital status at the time of surveys	80.2 (82.8–89.5 <i>)</i>	13.5 (10.5-17.2)	ŏı.4 (//.4–ŏ4.ŏ)	18.0 (15.2-22.0)
Cinete			7(0/710 700)	
Single	83.9 (80.5-86./)	16.1 (13.3–19.5)	/6.0 (/1.9–/9.8)	24.0 (20.2–28.1)

## Table 2 (continued)

Variables	Malawi		Burkina Faso	
	Not depressed % (CI)	Probably depressed % (CI)	Not depressed % (Cl)	Probably depressed % (CI)
Engaged	93.1 (63.7–99.0)	6.9 (1.0–36.3)	79.8 (61.5–90.7)	20.2 (9.3–38.5)
Married	92.0 (85.3–95.8)	8.0 (4.2–14.7)	88.5 (84.6–91.5)	11.5 (8.5–15.4)
Personal reaction to pregnancy				
Scared	84.4 (78.3–89.0)	15.6 (11.0–21.7)	77.9 (72.7–82.4)	22.1 (17.6–27.3)
Нарру	90.8 (85.7–94.2)	9.2 (5.8–14.3)	87.3 (83.3–90.4)	12.7 (9.6–16.7)
Disappointed	82.8 (78.1–86.8)	17.2 (13.3–21.9)	74.5 (67.4–80.4)	25.5 (19.6–32.6)
Birth status				
Ever given birth/has at least one child	84.7 (77.7–89.8)	15.3 (10.2–22.3)	73.9 (68.0–79.1)	26.1 (20.9–32.0)
Still pregnant/never gave birth	85.7 (82.3-88.4)	14.3 (11.6–17.7)	84.1 (80.9–86.9)	15.9 (13.1–19.1)
Exposed to IPV				
Yes	73.2 (70.8–80.9)	23.8 (19.1–29.2)	77.9 (74.4–81.0)	22.1 (19.0–25.6)
No	91.6 (88.4–93.9)	8.4 (6.1–11.6)	91.4 (86.6–94.7)	8.6 (5.3–13.4)
Neighborhood safety	5.40(5.30-5.49)	4.86(4.63-5.09)	4.73(4.61-4.84)	4.15(3.91-4.40)
Social assets and safety nets	4.10(3.94- 4.26)	4.01(3.63-4.39)	4.85(4.72–4.98)	4.32(4.04–4.59)

\* Mean differences were computed for "neighborhood safety" and "social assets and safety net

provision or child care support from the partner [29, 59]. In addition, denial of paternity sometimes exposes girls to rejection by their own families. The burden of childcare and resource provision rests on them entirely, even though they are unemployed. Paternity denial also represents betrayal, and studies suggest adolescent girls often contemplate abortion or killing their partners [29, 59]. In some of Burkina Faso's ethnic groups, such as the Mossi, where most of our participants belong, cultural norms prohibit an unmarried pregnant girl from living in the same household as her male family members, especially her father [60]. It is believed that if such girls are allowed to live with their male family members, it could result in deaths or other misfortunes happening to members of the family. For this reason, the girls who become pregnant must voluntarily or involuntarily join the partner responsible for her pregnancy, or alternatively her paternal aunt, who then acts as mediator. It is usually after her delivery that the girl can return to the parental home after rites of forgiveness. Girls dealing with these situations, combined with partner rejection, can lead to extreme distress, sadness and hopeless and losing interest in living.

In contrast, parental and partner support is protective against depression among pregnant and parenting adolescents. This finding is consistent with previous studies [19, 61]. Young mothers need lots of help to navigate their new role as parents. Childcare is tedious and requires human and material resources. Having access to adequate support could help build young mothers' resilience, determination, and self-esteem [62, 63]. Increasing the level of material and non-material support could be an important strategy for addressing poor mental health among pregnant and parenting adolescent girls.

Consistent with previous studies in the global north, our study also shows that community-level factors are associated with probable depression among pregnant and parenting adolescent girls [31, 32]. Unsafe neighborhoods are associated with an increased risk of depression. Fear of violence and actual exposure to violence in one's community could increase one's level of stress and anxiety. Neighborhoods with elevated levels of violence and crime are characterized by weak social ties, segregation, and the concentration of low-income earners [64]. However, it is important to interpret our findings with caution given that these are young girls' perceptions about safety in their community, and two people in the same community may differ in their perceptions.

Poverty has been linked to an elevated risk of depression. As such, the association between unsafe neighborhoods and depression could be bi-directional. We also found that having safety nets and support within the community was associated with lower odds of probable depression. This finding suggests that interventions to address depression among pregnant and parenting adolescents should be multidimensional and draw from community resources to be effective. 
 Table 3
 Multivariable logistic regression models showing individual, family and friend, and community-level factors associated with depression symptoms among pregnant and parenting girls

Variables	Malawi		Burkina Faso	
	Model 1	Model 2	Model 1	Model 2
Age				
13–16	1	1	1	1
17	1.12 (0.52–2.43)	2.22 (0.94-5.21)	1.37 (0.68–2.73)	1.86 (0.87-4.02)
18	1.17 (0.57-2.40)	2.84 (1.24-6.49)*	0.78 (0.41-1.48)	1.14 (0.55–2.37)
19	1.04 (0.51-2.10)	3.02 (1.33-6.87)**	0.69 (0.37-1.28)	1.24 (0.61–2.54)
Marital status				
Married	1	1	1	1
Separated	2.53 (1.31-4.88)**	1.62 (0.77-3.41)	2.01 (1.09-3.68)*	1.08 (0.48-2.42)
Single	2.03 (1.25-3.29)**	1.73 (0.89–3.36)	1.86 (1.24–2.79)**	0.76 (0.43-1.34)
Highest level of education				
Primary or less	1	1	1	1
Secondary	0.58 (0.35-0.95)*	0.47 (0.27-0.82)**	1.39 (1.004–1.93)*	1.34 (0.91–2.02)
Marital status at the time of pregnancy				
Single	1	1	1	
Engaged	0.37 (0.05-2.82)	0.43 (0.05-3.32)	0.75 (0.32–1.75)	1.40 (0.51-3.83)
Married	0.44 (0.21–0.89)*	0.76 (0.29–1.98)	0.42 (0.29–0.60)***	0.65 (0.30-1.43)
Personal reaction to pregnancy				
Scared	1	1	1	1
Нарру	0.53 (0.28-1.01)	0.89 (0.39-2.04)	0.51 (0.34–0.75)***	1.01 (0.54–1.89)
Disappointed	1.12 (0.68–1.84)	1.00 (0.58–1.73)	1.26 (0.84–1.89)	0.79 (0.49–1.26)
Birth status				
No birth/still pregnant	1	1	1	1
Ever birthed/has at least one child	0.94 (0.56–1.57)	0.60 (0.33-1.10)	0.55 (0.39–0.76)***	0.65 (0.44–0.94)*
Exposed to IPV				
No	1	1	1	1
Yes	3.44 (2.19–5.39)***	2.87 (1.77-4.66)***	2.72 (1.64-4.49)***	2.17 (1.23-3.82)**
Parent's reaction				
Нарру	1	1	1	1
Upset	2.60 (1.30-5.20)**	1.38 (0.45-4.21)	2.30 (1.59–3.32)***	0.67 (0.29–1.57)
Neutral	1.95 (0.86-4.45)	0.84 (0.27-2.57)	2.47 (1.42-4.29)**	0.88 (0.37-2.07)
Friend's reaction				
Нарру	1	1	1	1
Upset	2.15 (1.01-4.58)*	1.12 (0.40-3.17)	4.34 (2.55–7.38)***	1.99 (0.99–3.99)
Neutral	1.83 (0.90-3.72)	0.90 (0.34–2.37)	2.12 (1.44-3.09)***	1.09 (0.64–1.85)
Partner's reaction				
Нарру	1	1	1	1
Denied	2.13 (1.14-4.00)*	1.09 (0.44–2.72)	5.22 (3.18-8.57)***	3.14 (1.34–7.11)**
Upset	0.99 (0.51-1.93)	0.58 (0.26-1.27)	2.76 (1.75-4.35)***	1.52 (0.83–2.79)
Neutral	1.45 (0.85–2.48)	1.16 (0.60–2.25)	1.31 (0.72–2.38)	0.78 (0.38–1.59)
Parental support				
Good	1	1	1	1
Fair	2.37 (1.44–3.93)***	1.95 (1.08–3.52)*	1.47 (1.002-2.17)*	1.19 (0.75–1.89)
No/Poor	1.83 (1.03–3.24)*	1.59 (0.82–3.10)	3.25 (2.12–4.99)***	2.08 (1.22–3.55)**
Friends support	. ,	. ,	. ,	. ,
Good	1	1	1	1
Fair	2.32 (1.13-4.76)*	1.36 (0.61–3.07)	1.62 (1.01-2.60)*	1.04 (0.61–1.77)
No/poor	2.11 (1.12–3.96)*	1.17 (0.54–2.55)	2.94 (1.83–4.74)***	1.49 (0.85–2.64)

Variables	Malawi	Malawi		Burkina Faso	
	Model 1	Model 2	Model 1	Model 2	
Partner support					
Good	1	1	1	1	
Fair	2.34 (1.34-4.09)**	1.23 (0.64–2.37)	2.46 (1.61-3.74)***	1.78 (1.02–3.09)*	
No/poor	2.43 (1.47-4.02)***	1.37 (0.68–2.76)	3.38 (2.12–5.39)***	1.21 (0.58–2.52)	
Neighborhood safety	0.71 (0.60–0.83)***	0.74 (0.61–0.89)**	0.82 (0.75-0.90)***	0.81 (0.73-0.90)***	
Social assets and safety nets	0.98 (0.88–1.07)	0.94 (0.82–1.06)	0.87 (0.80-0.94)***	0.87 (0.78–0.96)**	

Exponentiated coefficients; 95% confidence intervals in brackets \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001

## **Limitation and strengths**

Our use of the socio-ecological model to explain the factors associated with depression is an important strength of this study. Also, our study drew data from pregnant and parenting adolescent girls in two African countries and adds to the body of evolving literature on the topic. However, our study is not without some limitations. Our data is limited to only one main city and environs in the country and is not nationally representative even though it offers important insights into the problem. Also, the evaluation of depressive illness is often carried out within clinical settings by healthcare workers. Because health professionals are much more trusted and can help address these mental health challenges, the information provided by PPA girls in such clinical settings is likely more reliable and accurate. Self-reporting of depression symptoms is subjected to social desirability bias. Since we used enumerators to administer the screening tool, it is possible that our study has underreported the prevalence of depression. Under-reporting of PPA girls who had had an abortion but are not currently pregnant or with a child during the listing stage is likely, given that abortion is highly stigmatized in the study settings.

## Conclusion

This study evaluated the prevalence of probable depression among pregnant and parenting adolescents in Burkina Faso and Malawi. It used the socio-ecological framework to identify factors associated with depressive symptoms. Probable depression symptoms are common among pregnant and parenting adolescents, suggesting the need to regularly screen them for depression during antenatal and postnatal checks. Factors associated with depression among pregnant and parenting girls operate at multiple levels suggesting a need for multilevel interventions that address all areas of vulnerabilities. Also, factors associated with probable depression slightly differ between our two study settings, suggesting a need for context-specific intervention.

## Abbreviations

EA Enumeration area

PPA Pregnant and parenting adolescent

#### Author contributions

AIA, EC, NS, NG, AMT, AKI, AM, CT, EOO, BAU and CWK made substantial contributions to the conception, design of the study, and acquisition of data. AIA conducted the data analysis. AIA, EC and NG wrote the first draft. All authors revised the draft critically for important intellectual content and approved the study for publication. All authors read and approved the final manuscript.

### Funding

This research was made possible through funding to the African Population and Health Research Center from the Swedish International Development Cooperation Agency for the Challenging the Politics of Social Exclusion project (Sida Contribution No. 12103). CK's writing time was partially covered by a grant from the International Development Research Centre (IDRC) for the Action to empower adolescent mothers in Burkina Faso and Malawi to improve their sexual and reproductive health project (Grant No. 109813–001). The views expressed herein do not necessarily represent those of IDRC or its Board of Governors or Sida.

#### Availability of data and materials

Data will be made available on reasonable request to the corresponding author.

### Declarations

### Ethics approval and consent to particpate

The University of Malawi Research Ethics Committee (UNIMAREC) and Burkina Faso's Ministry of Health Ethics Committee for Health Research approved the study, and we observed all guidelines for conducting research with human subjects. Research assistants were trained on research ethics before fieldwork. Voluntary informed consent was sought from all participants. For unmarried and pregnant minors, consent from parents and guardians was obtained while the minors themselves provided assent. We anonymized all the data to protect participants' privacy and confidentiality.

### **Consent for publication**

Not applicable.

#### **Competing interests**

All authors have no competing interests to declare.

### Author details

<sup>1</sup> Sexual, Reproductive, Maternal, New-born, Child and Adolescent Health (SRMNCAH) Unit, African Population and Health Research Center, Manga Close, Nairobi, Kenya. <sup>2</sup>Centre for Social Research, University of Malawi, P. O. Box 280, Zomba, Malawi. <sup>3</sup>Institut Supérieur des Sciences de La Population, Université Joseph Ki-Zerbo, B.P. 7118, Ouagadougou 03, Burkina Faso. <sup>4</sup>Dalla Lana School of Public Health, University of Toronto, 155 College St Room 500, Toronto, ON M5T 3M7, Canada.

### Received: 8 June 2022 Accepted: 23 February 2023 Published online: 07 March 2023

### References

- Gore FM, Bloem PJ, Patton GC, Ferguson J, Joseph V, Coffey C, et al. Global burden of disease in young people aged 10–24 years: a systematic analysis. Lancet. 2011;377(9783):2093–102.
- Kessler RC, Angermeyer M, Anthony JC, De Graaf R, Demyttenaere K, Gasquet I, et al. Lifetime prevalence and age-of-onset distributions of mental disorders in the World Health Organization's World Mental Health Survey Initiative. World Psychiatry. 2007;6(3):168.
- WHO. Adolescent mental health. Geneva: World Health Organisation; 2021. https://www.who.int/news-room/fact-sheets/detail/adolescentmental-health. Accessed 20 April 2022.
- Dahmen B, Konrad K, Jahnen L, Herpertz-Dahlmann B, Firk C. Mental health of teenage mothers: impact on the next generation. Nervenarzt. 2019;90(3):243–50.
- Hodgkinson S, Beers L, Southammakosane C, Lewin A. Addressing the mental health needs of pregnant and parenting adolescents. Pediatrics. 2014;133(1):114–22.
- WHO. Child and adolescent mental health. Geneva: WHO; 2019. https:// www.who.int/mental\_health/maternal-child/child\_adolescent/en/.
- Shiferaw G, Bacha L, Tsegaye D. Prevalence of depression and its associated factors among orphan children in orphanages in Ilu Abba Bor Zone, South West Ethiopia. Psychiatry J. 2018;2018:7.
- Recto P, Champion JD. Psychosocial risk factors for perinatal depression among female adolescents: a systematic review. Issues Ment Health Nurs. 2017;38(8):633–42.
- 9. SmithBattle L, Freed P. Teen mothers' mental health. MCN Am J Maternal Child Nurs. 2016;41(1):31–6.
- 10. Dinwiddie KJ, Schillerstrom TL, Schillerstrom JE. Postpartum depression in adolescent mothers. J Psychosom Obstet Gynecol. 2018;39(3):168–75.
- 11. Laurenzi CA, Gordon S, Abrahams N, du Toit S, Bradshaw M, Brand A, et al. Psychosocial interventions targeting mental health in pregnant adolescents and adolescent parents: a systematic review. Reprod Health. 2020;17:1–15.
- Barnet B, Duggan AK, Devoe M, Burrell L. The effect of volunteer home visitation for adolescent mothers on parenting and mental health outcomes: a randomized trial. Arch Pediatr Adolesc Med. 2002;156(12):1216–22.
- Roberts KJ, Smith C, Cluver L, Toska E, Sherr L. Understanding mental health in the context of adolescent pregnancy and HIV in Sub-Saharan Africa: a systematic review identifying a critical evidence gap. AIDS Behav. 2021. https://doi.org/10.1007/s10461-020-03138-z.
- Boyes ME, Cluver LD, Meinck F, Casale M, Newnham E. Mental health in South African adolescents living with HIV: correlates of internalising and externalising symptoms. AIDS Care. 2019;31(1):95–104.
- Cluver LD, Sherr L, Toska E, Zhou S, Mellins C-A, Omigbodun O, et al. From surviving to thriving: integrating mental health care into HIV community and family services for adolescents living with HIV. Lancet Child Adolescent Health. 2022. https://doi.org/10.1016/S2352-4642(22)00101-8.
- West N, Schwartz S, Mudavanhu M, Hanrahan C, France H, Nel J, et al. Mental health in South African adolescents living with HIV. AIDS Care. 2019;31(1):117–24.
- Woollett N, Cluver L, Bandeira M, Brahmbhatt H. Identifying risks for mental health problems in HIV positive adolescents accessing HIV treatment in Johannesburg. J Child Adolesc Ment Health. 2017;29(1):11–26.
- Laurenzi CA, Gordon S, Abrahams N, Du Toit S, Bradshaw M, Brand A, et al. Psychosocial interventions targeting mental health in pregnant

adolescents and adolescent parents: a systematic review. Reprod Health. 2020;17(1):1–15.

- Osok J, Kigamwa P, Huang K-Y, Grote N, Kumar M. Adversities and mental health needs of pregnant adolescents in Kenya: identifying interpersonal, practical, and cultural barriers to care. BMC Womens Health. 2018;18(1):96.
- Field S, Honikman S, Abrahams Z. Adolescent mothers: A qualitative study on barriers and facilitators to mental health in a low-resource setting in Cape Town, South Africa. Afr J Prim Health Care Fam Med. 2020;12(1):1–9.
- Buzi RS, Smith PB, Kozinetz CA, Peskin MF, Wiemann CM. A socioecological framework to assessing depression among pregnant teens. Matern Child Health J. 2015;19(10):2187–94.
- Gu J, Lau JT, Li M, Li H, Gao Q, Feng X, et al. Socio-ecological factors associated with depression, suicidal ideation and suicidal attempt among female injection drug users who are sex workers in China. Drug Alcohol Depend. 2014;144:102–10.
- 23. Ihekweazu EC. Predictors of Depression using the Socioecological Model. New Haven: Southern Connecticut State University; 2019.
- Poleshuck E, Perez-Diaz W, Wittink M, ReQua M, Harrington A, Katz J, et al. Resilience in the midst of chaos: socioecological model applied to women with depressive symptoms and socioeconomic disadvantage. J Community Psychol. 2019;47(5):1000–13.
- Ajayi Al, Akpan W, Goon DT, Nwokocha EE, Adeniyi OV. Tough love: sociocultural explanations for deadly abortion choices among Nigerian undergraduate students: health. Afr J Phys Act Health Sci. 2016;22(31):711–24.
- Ahinkorah BO, Onayemi OM, Seidu A-A, Awopegba OE, Ajayi AI. Association between girl-child marriage and intimate partner violence in Sub-Saharan Africa: insights from a multicountry analysis of demographic and health surveys. J Interpers Violence. 2021. https://doi.org/10.1177/08862 605211005139.
- Gillespie B, Allen H, Pritchard M, Soma-Pillay P, Balen J, Anumba D. Agency under constraint: adolescent accounts of pregnancy and motherhood in informal settlements in South Africa. Glob Public Health. 2021. https:// doi.org/10.1080/17441692.2021.1981974.
- Miller LE, Zamudio-Haas S, Otieno B, Amboka S, Odeny D, Agot I, et al. "We don't fear HIV. We just fear walking around pregnant": a qualitative analysis of adolescent sexuality and pregnancy stigma in informal settlements in Kisumu, Kenya. Stud Fam Plan. 2021;52(4):557–70.
- Coast E, Mwali MM, Isimbi R, Ngabonzima E, Pereznieto P, Buzby S, et al. 'If she's pregnant, then that means that her dreams fade away': exploring experiences of adolescent pregnancy and motherhood in Rwanda. Eur J Dev Res. 2021;33(5):1274–302.
- Boyle MH, Georgiades K, Duncan L, Wang L, Comeau J, Team OCHS. Poverty, neighbourhood antisocial behaviour, and children's mental health problems: findings from the 2014 Ontario Child Health Study. Can J Psychiatry. 2019;64(4):285–93.
- Leibbrand C, Hill H, Rowhani-Rahbar A, Rivara F. Invisible wounds: community exposure to gun homicides and adolescents' mental health and behavioral outcomes. SSM Popul health. 2020;12: 100689.
- Cuartas J, Leventhal T. Exposure to community violence and children's mental health: a quasi-experimental examination. Soc Sci Med. 2020;246: 112740.
- UNICEF. Early childbearing and teenage pregnancy rates by country New York: UNICEF; 2021. https://data.unicef.org/topic/child-health/adole scent-health/. Accessed 11 April 2022.
- Mmari K, Cooper D, Moreau C, Koenig L, Martinez M, Mafuta E, et al. The social context of early adolescents in the global early adolescent study. J Adolesc Health. 2021;69(1):S5–15.
- Levis B, Benedetti A, Thombs BD. Accuracy of Patient Health Questionnaire-9 (PHQ-9) for screening to detect major depression: individual participant data meta-analysis. Bmj. 2019;365:33.
- Manea L, Gilbody S, McMillan D. Optimal cut-off score for diagnosing depression with the patient health questionnaire (PHQ-9): a meta-analysis. CMAJ. 2012;184(3):E191–6.
- Costantini L, Pasquarella C, Odone A, Colucci ME, Costanza A, Serafini G, et al. Screening for depression in primary care with patient health questionnaire-9 (PHQ-9): a systematic review. J Affect Disord. 2021;279:473–83.
- 38. Anand P, Bhurji N, Williams N, Desai N. Comparison of PHQ-9 and PHQ-2 as screening tools for depression and school related

stress in inner city adolescents. J Prim Care Community Health. 2021;12:21501327211053750.

- Adewuya AO, Ola BA, Afolabi OO. Validity of the patient health questionnaire (PHQ-9) as a screening tool for depression amongst Nigerian university students. J Affect Disord. 2006;96(1–2):89–93.
- Nolan C, O'Donnell P, Desderius B, Mzombwe M, McNairy M, Peck R, et al. Depression screening in HIV-positive Tanzanian adults: comparing the PHQ-2, PHQ-9 and WHO-5 questionnaires. Glob Mental Health. 2018;5: e38.
- Makhubela M, Khumalo IP. Psychometric evaluation of the PHQ-9 in university students: factorial validity and measurement equivalence across three African countries. Curr Psychol. 2022;1–9:41.
- 42. Botha MN. Validation of the Patient Health Questionnaire (PHQ–9) in an African context. Potchefstroom: North-West University; 2011.
- StataCorp. Stata statistical software: release 15. College Station: StataCorp LLC; 2017.
- Mbawa M, Vidmar J, Chingwaru C, Chingwaru W. Understanding postpartum depression in adolescent mothers in Mashonaland Central and Bulawayo Provinces of Zimbabwe. Asian J Psychiatr. 2018;32:147–50.
- Govender D, Naidoo S, Taylor M. Antenatal and postpartum depression: prevalence and associated risk factors among adolescents' in KwaZulu-Natal, South Africa. Depress Res Treat. 2020. https://doi.org/10.1155/ 2020/5364521.
- Niyonsenga J, Mutabaruka J. Factors of postpartum depression among teen mothers in Rwanda: a cross-sectional study. J Psychosom Obstet Gynecol. 2020. https://doi.org/10.1080/0167482X.2020.1735340.
- Osok J, Kigamwa P, Vander Stoep A, Huang K-Y, Kumar M. Depression and its psychosocial risk factors in pregnant Kenyan adolescents: a crosssectional study in a community health centre of Nairobi. BMC Psychiatry. 2018;18(1):1–10.
- Kimbui E, Kuria M, Yator O, Kumar M. A cross-sectional study of depression with comorbid substance use dependency in pregnant adolescents from an informal settlement of Nairobi: drawing implications for treatment and prevention work. Ann Gen Psychiatry. 2018;17(1):1–15.
- 49. Eboreime E, Ezeokoli A, Adams K, Banke-Thomas A. Prioritizing the mental health needs of pregnant adolescents in sub-Saharan Africa. J Glob Health Neurol Psychiatry. 2022. https://doi.org/10.52872/001c.34646.
- Kuringe E, Materu J, Nyato D, Majani E, Ngeni F, Shao A, et al. Prevalence and correlates of depression and anxiety symptoms among out-ofschool adolescent girls and young women in Tanzania: a cross-sectional study. PLoS ONE. 2019;14(8): e0221053.
- Bjelland I, Krokstad S, Mykletun A, Dahl AA, Tell GS, Tambs K. Does a higher educational level protect against anxiety and depression? The HUNT study. Soc Sci Med. 2008;66(6):1334–45.
- Bauldry S. Variation in the protective effect of higher education against depression. Soc Mental health. 2015;5(2):145–61.
- Schaffnit SB, Urassa M, Lawson DW. "Child marriage" in context: exploring local attitudes towards early marriage in rural Tanzania. Sex Reprod Health Matters. 2019;27(1):93–105.
- Schaffnit SB, Hassan A, Urassa M, Lawson DW. Parent–offspring conflict unlikely to explain 'child marriage'in northwestern Tanzania. Nat Hum Behav. 2019;3(4):346–53.
- Brar SK, Beattie TSH, Abas M, Vansia D, Phanga T, Maseko B, et al. The relationship between intimate partner violence and probable depression among adolescent girls and young women in Lilongwe, Malawi. Glob Public Health. 2020;15(6):865–76.
- Okafor CN, Barnett W, Zar HJ, Nhapi R, Koen N, Shoptaw S, et al. Associations of emotional, physical, or sexual intimate partner violence and depression symptoms among South African women in a prospective cohort study. J Interpers Violence. 2021;36(9–10):5060–83.
- Esie P, Osypuk TL, Schuler SR, Bates LM. Intimate partner violence and depression in rural Bangladesh: accounting for violence severity in a high prevalence setting. SSM Popul Health. 2019;7: 100368.
- Yuan W, Hesketh T. Intimate partner violence and depression in women in China. J Interpers Violence. 2021;36(21–22):12016–40.
- Macleod C, Matebese S, Tsetse N. "I drank because i wanted to deal with the frustration": explaining alcohol consumption during pregnancy in a low-resource setting-women's, partners and family members' narratives. Soc Work. 2020;56(1):88–96.
- 60. Badini A. La représentation de la vie et de la mort chez les Mosé traditionnels de Haute-Volta. Villeneuve-d'Ascq: Université de Lille III; 1978.

- Kumar M, Huang K-Y, Othieno C, Wamalwa D, Madeghe B, Osok J, et al. Adolescent pregnancy and challenges in Kenyan context: perspectives from multiple community stakeholders. Glob Soc Welfare. 2018;5(1):11–27.
- 62. Pinkerton J, Dolan P. Family support, social capital, resilience and adolescent coping. Child Fam Soc Work. 2007;12(3):219–28.
- 63. Easterbrooks M, Kotake C, Raskin M, Bumgarner E. Patterns of depression among adolescent mothers: resilience related to father support and home visiting program. Am J Orthopsychiatry. 2016;86(1):61.
- 64. Gruebner O, Rapp MA, Adli M, Kluge U, Galea S, Heinz A. Cities and mental health. Dtsch Arztebl Int. 2017;114(8):121.

## **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

